Author's response to reviews

Title: Talk, Trust and Time: A Longitudinal Study Evaluating Knowledge Translation and Exchange Processes for Research on Violence Against Women

Authors:

C Nadine Wathen (nwathen@uwo.ca)
Shannon L. Sibbald (ssibbald@uwo.ca)
Susan M. Jack (jacksm@mcmaster.ca)
Harriet L. MacMillan (macmilnh@mcmaster.ca)

Version: 5 Date: 22 July 2011

Author's response to reviews:

The recent comment of the Associate Editor was:

"This paper can be recommended for acceptance. However could the authors pay a bit more attention in the abstract to describing the main themes identified from the research."

This has been addressed by re-working the abstract, especially the Results, to be more specific about the findings. Other sections of the Abstract have been edited for length. The new abstract is highlighted in grey to indicate changes.

Thanks.